Author's response to reviews

Title: Improving performance in medical practices through the extended use of electronic medical record systems: A survey of Canadian family physicians

Authors:

Louis Raymond (louis.raymond@uqtr.ca)
Guy Pare (guy.pare@hec.ca)
Ana Ortiz-de-Guinea (ana.ortiz-de-guinea@hec.ca)
Placide Poba-Nzaou (poba-nzaou.placide@uqam.ca)
Marie-Claude Trudel (marie-claude.trudel@hec.ca)
Josianne Marsan (josianne.marsan@sio.ulaval.ca)
Thomas Micheneau (thomas.micheneau@hec.ca)

Version: 4 Date: 24 February 2015

Author's response to reviews: see over
Dear Dr Pathak:

We thank you for the opportunity to revise and resubmit our work to BMC Medical Informatics & Decision Making. As suggested, our more round of proof reading was performed to correct all grammatical errors. This linguistic revision was performed by a certified professional.

Kind regards,

[Signature]

Chair in Information Technology in Health Care
Professor of Information Technology
HEC Montréal